# **Geophysical Research Letters**<sup>®</sup>

## **RESEARCH LETTER**

10.1029/2023GL107357

#### **Key Points:**

- We produced a new synthesis of subglacial boundaries for Greenland's geologic provinces from seismic, gravity, magnetic and topography data
- Three subglacial regions in central and northern Greenland cannot yet be reconciled with surface-exposed geologic provinces
- We find evidence for a large subglacial valley network that is not fully resolved by subglacial topography syntheses for Greenland

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

J. A. MacGregor, joseph.a.macgregor@nasa.gov

#### Citation:

MacGregor, J. A., Colgan, W. T., Paxman, G. J. G., Tinto, K. J., Csathó, B., Darbyshire, F. A., et al. (2024). Geologic provinces beneath the Greenland Ice Sheet constrained by geophysical data synthesis. *Geophysical Research Letters*, *51*, e2023GL107357. https://doi.org/10.1029/ 2023GL107357

Received 17 NOV 2023 Accepted 4 APR 2024

#### **Author Contributions:**

Conceptualization: Joseph A. MacGregor, William T. Colgan, Guy J. G. Paxman, Kirsty J. Tinto, Beáta Csathó, Fiona A. Darbyshire, Mark A. Fahnestock, Emma J. MacKie, Mathieu Morlighem, Olga V. Sergienko Data curation: Joseph A. MacGregor, William T. Colgan, Guy J. G. Paxman, Kirsty J. Tinto, Fiona A. Darbyshire Formal analysis: Joseph A. MacGregor, William T. Colgan, Guy J. G. Paxman,

Published 2024. This article is a U.S. Government work and is in the public domain in the USA. Geophysical Research Letters published by Wiley Periodicals LLC on behalf of American Geophysical Union.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

## **Geologic Provinces Beneath the Greenland Ice Sheet Constrained by Geophysical Data Synthesis**

Joseph A. MacGregor<sup>1</sup> <sup>(D)</sup>, William T. Colgan<sup>2</sup> <sup>(D)</sup>, Guy J. G. Paxman<sup>3</sup> <sup>(D)</sup>, Kirsty J. Tinto<sup>4</sup> <sup>(D)</sup>, Beáta Csathó<sup>5</sup> <sup>(D)</sup>, Fiona A. Darbyshire<sup>6,7</sup> <sup>(D)</sup>, Mark A. Fahnestock<sup>8</sup> <sup>(D)</sup>, Thomas F. Kokfelt<sup>2</sup>, Emma J. MacKie<sup>9</sup> <sup>(D)</sup>, Mathieu Morlighem<sup>10</sup> <sup>(D)</sup>, and Olga V. Sergienko<sup>11</sup> <sup>(D)</sup>

<sup>1</sup>Cryospheric Sciences Laboratory, NASA Goddard Space Flight Center, Greenbelt, MD, USA, <sup>2</sup>Geological Survey of Denmark and Greenland, Copenhagen, Denmark, <sup>3</sup>Department of Geography, Durham University, Durham, UK, <sup>4</sup>Lamont-Doherty Earth Observatory, Columbia University, Palisades, NY, USA, <sup>5</sup>Department of Geology, University at Buffalo, Buffalo, NY, USA, <sup>6</sup>Centre de recherche Geotop, Université du Québec à Montréal, Montréal, QC, Canada, <sup>7</sup>Département des sciences de la Terre et de l'atmosphère, Université du Québec à Montréal, Montréal, QC, Canada, <sup>8</sup>Geophysical Institute, University of Alaska Fairbanks, Fairbanks, AK, USA, <sup>9</sup>Department of Geological Sciences, University of Florida, Gainesville, FL, USA, <sup>10</sup>Department of Earth Sciences, Dartmouth College, Hanover, NH, USA, <sup>11</sup>Program in Atmospheric and Oceanic Sciences, Princeton University, Princeton, NJ, USA

**Abstract** Present understanding of Greenland's subglacial geology is derived mostly from interpolation of geologic mapping of its ice-free margins and unconstrained by geophysical data. Here we refine the extent of its geologic provinces by synthesizing geophysical constraints on subglacial geology from seismic, gravity, magnetic and topographic data. North of 72°N, no province clearly extends across the whole island, leaving three distinct subglacial regions yet to be reconciled with margin geology. Geophysically coherent anomalies and apparent province boundaries are adjacent to the onset of faster ice flow at both Petermann Glacier and the Northeast Greenland Ice Stream. Separately, based on their subaerial expression, dozens of unusually long, straight and sub-parallel subglacial valleys cross Greenland's interior and are not yet resolved by current syntheses of its subglacial topography.

**Plain Language Summary** The Greenland Ice Sheet obscures the rocks beneath 79% of Greenland. By necessity, scientists have relied mostly on studying the rocks exposed along Greenland's edge to understand the island's interior geology. We examine geophysical data from seismometers on the ground, satellites that measure Earth's gravity and magnetic fields and surface topography, and aircraft that measure those same properties and ice thickness. We draw a new map of Greenland's geology beneath the ice sheet by examining where those data show similar signals regarding the nature of the underlying rock, and where they could be related to mapped rock exposures. We also find evidence of some areas with geophysical expressions that are distinct from the rocks found at the island's edges. Some geologic structures, which are entirely covered by ice, may affect how ice flows from Greenland's vast interior toward its coast. Finally, we identify many valleys beneath the ice that are very long and often aligned with each other, but which are not yet fully captured in present maps of the topography beneath the ice.

#### 1. Introduction

The Greenland Ice Sheet (*GrIS*; not including peripheral ice masses) covers 1.1% of Earth's land surface and nearly 79% of Greenland itself. The ice sheet's large-scale flow pattern, its discharge into the surrounding oceans and the island's response to changing ice loading are all influenced by solid-Earth boundary conditions that are obscured by the GrIS and thus difficult to constrain. The solid-Earth boundary conditions that have received the most attention are its subglacial topography, geothermal heat flux, crustal geology and mantle viscosity (e.g., Adhikari et al., 2021; Alley et al., 2019; Braun et al., 2007; Colgan et al., 2022; Martos et al., 2018; Morlighem et al., 2017; Rezvanbehbahani et al., 2017; Rogozhina et al., 2016). These properties are potentially related, for example, a geologically young province with a rough topographic surface may have a higher geothermal heat flux and overlie a less viscous mantle, but these relations remain difficult to constrain precisely despite decades of investigation.

Of these properties, Greenland's subglacial geology is perhaps the most poorly constrained, despite its potential influence on ice flow (e.g., Winsborrow et al., 2010). Dawes (2009) (hereafter *D09*) produced the seminal map of

ADVANCING



Check for updates



Kirsty J. Tinto, Beáta Csathó, Fiona A. Darbyshire, Mark A. Fahnestock, Thomas F. Kokfelt

#### Funding acquisition: Joseph A. MacGregor

Investigation: Joseph A. MacGregor, William T. Colgan, Guy J. G. Paxman, Kirsty J. Tinto, Beáta Csathó, Mark A. Fahnestock, Thomas F. Kokfelt, Emma J. MacKie, Mathieu Morlighem, Olga V. Sergienko

Methodology: Joseph A. MacGregor, William T. Colgan, Guy J. G. Paxman, Kirsty J. Tinto, Beáta Csathó, Fiona A. Darbyshire, Mark A. Fahnestock, Thomas F. Kokfelt, Emma J. MacKie, Mathieu Morlighem, Olga V. Sergienko Project administration: Joseph

A. MacGregor

Resources: Joseph A. MacGregor Software: Joseph A. MacGregor, Mark A. Fahnestock

Supervision: Joseph A. MacGregor Validation: Joseph A. MacGregor, William T. Colgan, Guy J. G. Paxman, Mark A. Fahnestock, Thomas F. Kokfelt, Emma J. MacKie, Mathieu Morlighem, Olga V. Sergienko Visualization: Joseph A. MacGregor, William T. Colgan, Guy J. G. Paxman,

Kirsty J. Tinto, Mark A. Fahnestock, Thomas F. Kokfelt, Mathieu Morlighem, Olga V. Sergienko

Writing – original draft: Joseph A. MacGregor, William T. Colgan, Guy J. G. Paxman, Fiona A. Darbyshire Writing – review & editing: Joseph A. MacGregor, William T. Colgan, Guy J. G. Paxman, Kirsty J. Tinto, Beáta Csathó, Fiona A. Darbyshire, Mark A. Fahnestock, Thomas F. Kokfelt, Emma J. MacKie, Mathieu Morlighem, Olga V. Sergienko Greenland's subglacial geology that is still widely used for both interpretation of island-wide studies and global maps of geologic provinces (e.g., Alley et al., 2019; Darbyshire et al., 2018; Gowan et al., 2019; Hasterok et al., 2022; Jones et al., 2021; Maier et al., 2022; Martos et al., 2018; Mordret, 2018; Pourpoint et al., 2018; White et al., 2016). However, the boundaries between major provinces that D09 identified were typically unconstrained across several hundred kilometers between similar synchronous formations known only from Greenland's ice-free periphery. Contemporaneous knowledge of magnetic anomalies was considered only for southern Greenland and seismologic inferences received limited consideration. Subglacial topography was not considered, despite being directly identified as a potential constraint on subglacial geology, nor were gravity data. D09 acknowledged these limitations by describing their resulting map as "conjectural" and mapping of subglacial geology as "the next major challenge for Greenland mapping."

Since D09, the extent of marginal geologic mapping, seismic imaging, aerogeophysical surveys, satellite mapping of gravity and magnetic anomalies, and surface properties across Greenland have improved substantially. An opportunity exists to improve our understanding of Greenland's subglacial geology by synthesizing this new wealth of pertinent data and associated analyses. Here we map major boundaries in these data sets and reconcile these geophysical constraints with marginal geologic exposures and samples from beneath and within deep ice cores. We consider the implications of this new map of Greenland's geologic provinces for interpretation of Greenland's geologic history.

### 2. Data and Methods

#### 2.1. Province Delineation

We evaluated 19 geologic and geophysical data sets for this study to delineate major subglacial geologic provinces beneath the GrIS (Table S1 in Supporting Information S1). Most are evaluated as is with only minor standard corrections applied. A primary challenge in reconciling these data sets lies in their differing depth sensitivities, resolutions, interpolations and interpretations. The spatial resolution, precision and accuracy of each data set are discussed in detail in the cited references; deeper-sensing or satellite-derived data sets typically have coarser resolution. For simplicity, we only consider airborne data collected by NASA's Operation IceBridge and predecessor NASA surveys (MacGregor et al., 2021). Except for nearby subaerially exposed rocks, the Greenland Ice Sheet Project 2 (*GISP2*) bedrock sample and to lesser extent ice-core basal ice lithologies, all data sets are indirect indicators of subglacial geology, but none are uniquely diagnostic of a particular subglacial geologic province. However, combinations thereof have been used regularly to diagnose subglacial geology in Greenland and Antarctica (e.g., Brozena, 1995; Fahnestock et al., 2001; Jordan et al., 2023; MacGregor et al., 2019; Paxman, Austermann, & Tinto, 2021; Paxman, Tinto, & Austermann, 2021; Paxman et al., 2019; Rogozhina et al., 2016; Tinto et al., 2019).

We generally selected the most recent, comprehensive and accessible data set for each relevant method. The numerous existing inferences of Greenland crustal properties from seismic tomography are particularly challenging to reconcile, so here we consider only one recent study that applied state-of-the-art methods to a recent seismic data catalog to explore Greenland's crustal velocity structure (Darbyshire et al., 2018). While some geophysical inversions can coarsely resolve three-dimensional crustal structure, here we focus on inferences relevant to the delineation of apparent geologic provinces at their outcrop, that is, as they might be exposed subaerially if the GrIS and any surficial sediment cover were absent.

We first combine the D09 map of Greenland's subglacial geology and a more recent map of the island's subaerially exposed geology to delineate subaerial province extents (Pedersen et al., 2013; Watt, 2019). At a regional scale (~1:500,000), we then manually trace apparent boundaries at >10-km spacing for each geophysical field separately. Where possible, this delineation is guided by each field's relation to mapped subaerial province boundaries in ice-free regions. The subglacial boundaries are based on where the magnitude, azimuthal trend or regional-scale pattern of each field show significant discontinuities, indicating a possible change in subglacial geology (e.g., Paxman et al., 2019). Based on the apparent significance of each discontinuity, geophysical boundaries are assigned either low or high confidence. For subglacial topography, we trace one boundary set that considers the ice-free rebounded subglacial topography, and a second set based on the ice sheet's surface expression of present-day subglacial topography. The latter set is informed by the pattern of mapped surface lineations (Section 2.2).

We then manually delineate new subglacial geologic boundaries based on our geophysical boundaries, their spatial coherence across both their ice-free exposures and apparent subglacial extent, our confidence therein, and consensus among this study's authors. Where multiple margin-originating geophysical boundaries are aligned within ~50 km of each other, we identify the geologic boundary as high-confidence and assign it to either the highest-confidence geophysical boundary or the midpoint between several of them. We identify most apparent geologic boundaries that do not reach the ice-sheet margin as low-confidence. Where available data cannot be reconciled with known provinces, we identify the region as unresolved. Similarly, we do not assign subglacial geology beneath Greenland's smaller peripheral ice masses, because these are often poorly resolved by the geophysical data sets considered here and because several of them reside near known transitions in subaerial geology (e.g., Flade Isblink in far northeastern Greenland).

#### 2.2. Flow-Aware Hillshade

We introduce one new method: the calculation and interpretation of a "flow-aware" hillshade of a digital elevation model (*DEM*) of the ice surface. This method is motivated by the observation and supporting theory that an ice mass flowing over variable bed topography transfers a signature of that subglacial topography to its subaerial surface (Gudmundsson, 2003; Ng et al., 2018), a phenomenon that is not presently considered to infer bed topography in the ice-sheet interior (Morlighem et al., 2022). To calculate this hillshade, we first filter the measured flow direction across the GrIS interior (Joughin et al., 2016, 2017) toward the direction of the surface-elevation gradient instead, as measured by the Greenland Ice Mapping Project (GrIMP; Howat et al., 2022) and following MacGregor et al. (2022). We illuminate the GrIMP DEM using a standard hillshade algorithm (Greene et al., 2017), modified so that the illumination azimuth may vary for each pixel, which we select as 90° counterclockwise to the filtered flow direction. The advantage of the flow-aware hillshade is that it highlights surface-slope features while cognizant of the frame of reference most relevant to an ice mass (its flow direction). Its appearance is similar to the "residual surface elevation" method introduced by Cooper et al. (2019) but the flow-aware hillshade (hereafter simply *hillshade*) is easier to implement ice-sheet-wide.

The resulting hillshade highlights sub-kilometer-scale variability in GrIS subaerial surface slope (Figure 1). At this scale, variations in surface slope are related primarily to variability in subglacial topography, although this bed-to-surface transfer is complex and depends on the rate of basal sliding and rheology of the ice column (Ng et al., 2018). Contiguous surface slope features are often interpreted as subglacial valley networks in both Greenland and Antarctica (e.g., Ekholm et al., 1998; Livingstone et al., 2017; Ross et al., 2014; Scambos & Haran, 2002). Changes in surface slope may also be related to short length-scale variability in subglacial sliding (Cooper et al., 2019; Gudmundsson, 2003), but these are also detectable by surface-velocity observations and appear to be uncommon—shear margins notwithstanding. As an indirect but independent constraint on ice-thickness-scale subglacial topography, the hillshade is especially valuable for interpreting gaps between radar-sounding surveys that are widespread across the GrIS interior (Morlighem et al., 2022).

The GrIMP hillshade highlights innumerable subtle but coherent surface lineations that are associated with ice flow across subglacial valleys (Figures 1b and 1d) and mountains (Figure 1e), although their interpretation is complicated by the geometry of the radar–lineation intersections and their relation to the ice-flow direction. Some of these lineations are also visible in Moderate Resolution Imaging Spectroradiometer (*MODIS*) mosaics (Haran et al., 2013), but many more are evident in this GrIMP hillshade. We manually traced prominent surface lineations across the GrIS, focusing on those that were longer than 10 km, wider than 1 km, detectable in the interior, or intersected radar-sounding surveys. This tracing is non-exhaustive but representative of the dominant lineation patterns.

#### 3. Results

Figure 2 shows the traced boundaries for each geophysical field. For most province boundaries south of  $\sim$ 70°N, there is general agreement on their location (boundaries within  $\sim$ 100 km of each other), except for boundaries in the depth to the Mohorovičić discontinuity, which are generally meridionally oriented and rarely coincide with other boundaries. Farther north, some boundaries roughly coincide across multiple fields, such as in far northwestern and eastern central Greenland, but more common is a complex set of intersecting boundaries. This complexity favors our synthesis approach, rather than a focus on one geophysical field as in single-method studies.



## **Geophysical Research Letters**



**Figure 1.** Flow-aware hillshade of the GrIS derived from the GrIMP DEM. (a) Map of the whole island with manually traced lineations (Figure S1 in Supporting Information S1 shows uninterpreted hillshade). (b–g) Zoom-ins of selected regions with bed-elevation anomaly  $\Delta z_b$  (subtracted from 5-km running mean) shown where  $|\Delta z_b| \ge 50$  m in all 1993–2019 NASA radar-sounding data (CReSIS, 2024; thin white lines). The scale is the same for all six panels. Selected 50-km-long segments of radar-sounding tracks (thick white lines) are shown in the rightmost column, along with the surface-elevation anomaly. Cyan arrows on maps show mean ice-flow direction along those segments; the mean ice thickness in meters ( $\overline{H}$ ) for each segment is shown in the lower right of each profile.

Figure 3 shows our synthesis interpretation of apparent subglacial geologic provinces, Table S2 in Supporting Information S1 summarizes their geophysical expressions and Figure S2 in Supporting Information S1 compares each field's boundaries with our synthesis. We delineate 12 partly subglacial provinces; those whose extent differ substantially from D09 are discussed below (Figure S3 in Supporting Information S1). Five additional fully subglacial regions are discussed, four of which are delineated in this study.

For the Inglefield Orogen, we do not find evidence for a dissected eastward extension from Inglefield Land ( $\sim$ 78.5°N, 68°W) across to northeastern Greenland, but rather multiple high-confidence geophysical boundaries at  $\sim$ 50°W. Following D09 and Nutman et al. (2019), we distinguish the Victoria Fjord Terrane exposed at the head of Victoria Fjord ( $\sim$ 81.6°N, 45.5°W) from the younger Rae Craton. For this terrane, we find some geophysical evidence for narrower eastern and western subglacial extensions and a more southerly extension from Victoria Fjord. For the main exposure of the North Atlantic Igneous Province along the Blosseville Coast ( $\sim$ 69°N, 25°W), multiple geophysical boundaries indicate a limited extension inland, as is the case for the Nuussuaq Basin in western Greenland. The geophysical expression of the predominant Rae Craton is varied, but our mapped boundaries suggest that its inland extent from the northeastern coast is generally  $\sim$ 100 km west of the Petermann Megacanyon (Figure 3b; Bamber et al., 2013), although it may extend farther inland into northern central Greenland. The inland boundary of the Rinkian Orogen is approximately constrained along the northwestern coast, while the better constrained extents of the North Atlantic Craton, the Nagssugtoqidian, Ketilidian and Caledonian orogens are similar to D09. Conversely, we interpret the Independence Fjord Basin to extend farther inland to the west, and it appears to underlie the mid-section of the Northeast Greenland Ice Stream (*NEGIS*; Figures 3b and 3d).

We outline three larger regions whose subglacial geophysical expressions cannot yet be reconciled with the inland extension of margin-exposed provinces. In the central interior of northern Greenland, we outline a large region with relatively indistinct and lower-magnitude anomalies for most fields, which is bounded by known provinces



## **Geophysical Research Letters**



**Figure 2.** Geophysical constraints on GrIS subglacial geology and their traced boundaries. (a, b) Depth to Mohorovičić discontinuity and S-wave speed anomaly at 10 km depth, respectively, from seismic tomography (Darbyshire et al., 2018). (c) Bouguer gravity anomaly (Forsberg & Jensen, 2015). (d) Earth Magnetic Anomaly Grid (Meyer et al., 2017). For panels C and D, Operation IceBridge airborne gravity and magnetic data are overlain (Cochran & Bell, 2016; Cochran et al., 2014; K. Tinto, 2020). (e) Isostatically rebounded BedMachine v5 topography assuming ice-free conditions (Morlighem et al., 2017, 2022; Paxman et al., 2022a, 2022b). (f) GrIMP hillshade with surface-exposed provinces and major faults (Harrison et al., 2011; Watt, 2019). Colors for provinces follow Figure 3a. (g) All traced geophysical boundaries. Background map is MODIS Mosaic of Greenland (Haran et al., 2013).

with more distinct and widespread geophysical expressions (e.g., gravity and magnetic highs to its northwest and northeast). GISP2 is located near the apparent southeastern boundary of this region (Figure 3a), and its granitic bedrock appears consistent with the Rinkian Orogen (Weis et al., 1997). However, given limited geophysical similarity with the outcrops of that province or the regionally predominant Rae Craton, we opt not to ascribe this region of unknown geologic age and uncertain boundaries to any known province, and instead denote it generically as "Central Region A."

Directly southeast of Central Region A, we outline a separate interior region with distinct gravity and magnetic anomalies (in some cases co-located), along with seismically slow crust, and rough subglacial and surface topography. This combination of properties suggests a volcanic origin and unknown age but likely younger than Central Region A (e.g., Fahnestock et al., 2001). This province is located just south of the onset region of the NEGIS, and ice flows across its northern boundary toward the southeastern flank of the NEGIS (Figure 3d). Both Rogozhina et al. (2016) and Jones et al. (2021) have previously inferred higher geothermal heat flux in the vicinity of this region as a potential contributor to the anomalous flow of NEGIS. This approximate region was also previously described as the "Central Greenland Magmatic Province" by Brozena (1995), based on its aero-geophysical properties, but we designate it agnostically and generically as "Central Region B."



## **Geophysical Research Letters**



**Figure 3.** (a) New synthesis of apparent subglacial geologic provinces beneath the GrIS colored by age (Table S2 in Supporting Information S1; Cohen et al., 2013 updated 2023 at https://stratigraphy.org; Watt, 2019). Bedrock and basal sediment lithology and ages from three interior boreholes and ice cores also shown (Blard et al., 2023; Christ et al., 2023; Weis et al., 1997). (b) Major glaciological and subglacial features discussed with surface speed overlain (Joughin et al., 2016, 2017): 1. Candidate Iceland hotspot track (Martos et al., 2018); 2. Petermann Megacanyon (Paxman, Tinto, & Austermann, 2021); 3. NEGIS; and 4. Traced surface lineations (Figure 1a), emphasizing those  $\geq 100$  km long and  $\leq 1.1$  sinuosity. Inset rose diagram shows all lineation azimuths, with predominant 30–60° range emphasized. (c, d) Zoom-ins on the onsets of Petermann Glacier and NEGIS, respectively, with GrIMP hillshade overlain.

A third region borders both Central Regions A and B in eastern Greenland. Both its observed and apparent subglacial topography are notably rougher than those regions, and it shares few other geophysical characteristics with those regions or the Caledonian Orogen to its east (Figure 2). The simplest geologic explanation for this region may be that it is part of the Rae Craton (e.g., Thrane, 2002), but it is sufficiently distinct geophysically that we designate this third ambiguous region as "Central Region C."

Tinto et al. (2015) identified a prominent ellipsoidal gravity, magnetic and smooth topographic high east of Petermann Glacier, northwestern Greenland (Figure 3c), which they interpreted as an "igneous body." It is clearly bounded on its western flank by the onset region of faster ice flow of Petermann Glacier, and along its northern and eastern flanks by the Petermann Megacanyon (Figure 3c). This  $\sim$ 4,000 km<sup>2</sup> region is not large enough to constitute a separate province, so we nominally interpret it as part of the Inglefield Orogen and denote it generically as the "Petermann Anomaly." Based on basalt and porphyry erratics found in northwestern Greenland, Dawes et al. (2000) hypothesized the presence of an unexposed volcanic subglacial province beneath the northern GrIS. This putative province was outlined by D09 and—coincidentally—the Petermann

Anomaly is at the center of that outline and nearly entirely within it (Figure S3 in Supporting Information S1). If this feature is indeed an igneous body and if ice flow over it was diverted farther west during a more extensive Pleistocene glaciation, then it is conceivably the source of those volcanic erratics and likely somewhat younger than the Inglefield Orogen. Separately, we do not find clear geophysical evidence to bound another possible unexposed province identified by D09 in northwestern Greenland that could explain the presence of redbed erratics found along the coast of Baffin Bay (Figure S3 in Supporting Information S1). For completeness, we still include this putative yet unlocated subglacial province in Table S2 in Supporting Information S1.

We traced 1062 surface lineations within the GrIMP hillshade (Figures 1a and 3b). Where they intersect radarsounding surveys, these lineations are mostly related to sparsely surveyed subglacial valleys. Many of these lineations connect farther downstream with the better-surveyed deeply incised troughs beneath most outlet glaciers around the GrIS periphery (Morlighem et al., 2017); along the northwestern coast facing Baffin Bay, several of these valleys appear to be >300 km long. Networks of these lineations are typically subparallel to each other across hundreds of kilometers, an unusual configuration that implicates past crustal tectonics or subglacial incision as a control on their morphology (Rose et al., 2014). Near the margin, the azimuths of these subglacial valley networks are often similar to those of previously mapped major faults in adjacent ice-free forelands (Figure 2f), further suggesting a relation to subglacial geology.

#### 4. Discussion

A substantive change between our mapping of subglacial geologic provinces and D09 is that we do not find clear geophysical support for a dissected subglacial connection between the North Atlantic Igneous Province and the Nuussuaq Basin (Figure S3 in Supporting Information S1). If Central Region B is indeed magmatic, as hypothesized by Brozena (1995), it extends too far north to connect those peripheral provinces in a simple way. If there is a connection between those provinces and this unresolved region—perhaps associated with the Iceland hotspot for which numerous tracks have been hypothesized (cf., Martos et al., 2018)—then this region may support existing interpretations of a thinner lithosphere in this region due to thermal erosion by the hotspot (Nielsen et al., 2002; Steinberger et al., 2019). The hotspot track suggested by Martos et al. (2018), which was based primarily on magnetic data, trends farther north than most other tracks and may have been spatially coincident with Central Region B at ~60 Ma. Together, their hotspot track and the distinct geophysical properties of Central Region B support the existing hypothesis that the transit of the Iceland hotspot beneath north-central Greenland produced anomalous lithospheric heat that influences present crustal properties (Alley et al., 2019).

A comprehensive evaluation of the relation between radar-observed subglacial topography and its surface expression is beyond the scope of this study. However, the remarkable similarity between simple running-mean anomalies of the radar-measured subglacial topography and surface elevation indicates that a non-negligible relation between kilometer-scale subglacial and subaerial topography is common across the GrIS (Figure 1). This relation is qualitatively consistent with that predicted from linearized small-perturbation theory for bed-to-surface transfer (Gudmundsson, 2003; Ng et al., 2018), that is, subglacial topographic perturbations that are sufficiently wide along-flow relative to local ice thickness commonly induce detectable surface expressions. That linearized theory is potentially exploitable to resolve subglacial topography at finer resolution (Ockenden et al., 2022, 2023), but more physically complete modeling has demonstrated that three-dimensional effects and non-linear ice rheology further complicate its interpretation (e.g., Sergienko, 2012). Regardless, it is clear that numerous subglacial topographic features across the GrIS interior—primarily subglacial valleys—are poorly resolved by existing surveys and interpolations.

Adjacent to the highlands of eastern Greenland, the ice surface of Central Region C and northeastern Central Region B suggests an underlying alpine-style subglacial geomorphology reminiscent of the Gamburtsev Subglacial Mountains in Antarctica (e.g., Figure 1e; Lea et al., 2023; Rose et al., 2013). However, the rough bed topography of this region is not yet well captured in current syntheses as it has only been sparsely surveyed so far (Morlighem et al., 2022). Traced surface lineations tend to be shorter there than elsewhere across the GrIS and cover a broad azimuthal range (Figure 3b). We suggest that the existing hypothesis that this region hosted part of the nascent GrIS during the Pliocene and models indicating that it remains ice-covered until the final phase of deglaciation are consistent with its apparently well-preserved subglacial morphology (Aschwanden et al., 2019; Bierman et al., 2016; Paxman et al., 2024).

Across the rest of Greenland, 6% (n = 64) of the traced surface lineations are both longer than 100 km and have a sinuosity less than 1.1 (Figure 3b), and all appear to overlie subglacial valleys (Figure 1). The presence of these remarkably long and straight valleys suggests low variability in bedrock resistance to channel incision across the low topographic relief of Greenland's interior (Lazarus & Constantine, 2013), a pattern that may be consistent with the predominance of Archean and Proterozoic provinces across the island. Further, there appear to be two roughly orthogonal preferential directions for many of the features/valleys, with the southwest–northeast azimuth being predominant (30–60°; Figure 3b). This pattern suggests that Greenland's tectonic history generated preferential pathways of lithological weakness for subsequent fluvial or subglacial erosion to exploit (e.g., Dühnforth et al., 2010). The converse scenario—whereby high-competence bedrock redirects valley incision—appears to be valid for at least part of Greenland, that is, the Petermann Megacanyon steers around the relatively smooth and presumably competent Petermann Anomaly (Figure 3c; Paxman, Tinto, & Austermann, 2021).

Whether this subset of long, linear valleys formed by fluvial or subglacial processes remains unclear. Where they are intersected by airborne radar sounding data, the cross-sectional morphology of these valleys shows a range of forms resembling both glacial and fluvial valleys elsewhere in the Northern Hemisphere, although definitively U-shaped cross-profiles are uncommon in the Greenlandic interior (Paxman, 2023). A fluvial origin is most likely given low long-term glacial erosion rates in the vicinity of GISP2 in central Greenland (Bierman et al., 2014), although subglacial meltwater may have also contributed to valley incision where the bed is thawed (e.g., Kirkham et al., 2022).

#### 5. Conclusions

We improved knowledge of Greenland's subglacial geology by synthesizing multiple island-wide geologic, seismic, gravity, magnetic and topographic data sets that have been developed since Dawes (2009). This geophysically driven synthesis clarifies the apparent extents of Greenland's geologic provinces, but it still indicates that a greater variety of provinces exist in northern Greenland than southern Greenland. Three regions in central and northern Greenland possess sufficiently distinct geophysical expressions that they cannot yet be assigned to known provinces. While this new map is an inevitably incomplete representation of Greenland's subglacial geology, it provides a modernized framework for combined interpretation of the subglacial and solid-Earth properties of Greenland. Additionally, our observation of an extensive subglacial valley network beneath the GrIS, with many long and sub-parallel valleys, suggests that tectonic inheritance may have had a larger influence on Greenland's subglacial topography than previously recognized.

This new synthesis of Greenland's subglacial geology has value for several potential applications. Foremost is the modern context it provides for new applications of existing methods (e.g., interpretation of new seismic, gravity or magnetic data sets), evaluation of aerogeophysical data sets beyond those we considered, or tracing the provenance of offshore sediments (e.g., White et al., 2016). Another application could be exploring relations between these apparent province boundaries and other subglacial properties of geophysical, glaciological, geomorphological or geochemical interest, such as geothermal heat flow (Jones et al., 2021), bedrock erodibility (Campforts et al., 2020), basal friction (Maier et al., 2022), drainage history (Jess et al., 2020; Keisling et al., 2020), or isotope geochemistry (Briner et al., 2022; Colville et al., 2011). While our results were based on conventional manual delineation of province boundaries and interpretation, machine learning techniques could also be applied to reveal such structures with reduced influence from expert biases (e.g., Colgan et al., 2022; Li et al., 2022; Rezvanbehbahani et al., 2017).

#### **Data Availability Statement**

Most data sets used in this study are openly archived (Table S1 in Supporting Information S1), and some through QGreenland v3.0.0 (Fisher et al., 2023). Version 2 of the Grid files of the Total isostatic response to the complete unloading of the GrIS is publicly archived (Paxman et al., 2022b). All data sets that were previously unavailable (seismic tomography from Darbyshire et al., 2018) and those we generated (GrIMP flow-aware hillshade and associated code, traced surface lineations, traced boundaries in geophysical fields and the new subglacial geologic province map), are publicly archived (MacGregor et al., 2024).

#### Acknowledgments

We thank the NASA HO/GSFC Internal Scientist Funding Model and the NASA/ HQ Earth Surface and Interior Program for supporting JAM. We thank the GrIMP and QGreenland projects, the NASA MEaSUREs program, the National Snow and Ice Data Center, the NSF Arctic Data Center and Zenodo for archiving and distribution of some of the data sets examined in this study. We thank A. M. Chartrand, W. Chu, P. R. Dawes, D. Felikson, C. A. Greene, J. R. Hopper, N. B. Karlsson, M. Pourpoint, M. Studinger and L. Wang for valuable discussions. We thank Editor H. Rajaram and referees N. Ross and J. Briner for constructive evaluations

#### References

- Adhikari, S., Milne, G. A., Caron, L., Khan, S. A., Kjeldsen, K. K., Nilsson, J., et al. (2021). Decadal to centennial timescale mantle viscosity inferred from modern crustal uplift rates in Greenland. *Geophysical Research Letters*, 48(19), e2021GL094040. https://doi.org/10.1029/ 2021GL094040
- Alley, R. B., Pollard, D., Parizek, B. R., Anandakrishnan, S., Pourpoint, M., Stevens, N. T., et al. (2019). Possible role for tectonics in the evolving stability of the Greenland Ice Sheet. Journal of Geophysical Research: Earth Surface, 124(1), 97–115. https://doi.org/10.1029/2018JF004714
- Aschwanden, A., Fahnestock, M., Truffer, M., Brinkerhoff, D., Hock, R., Khroulev, C., et al. (2019). Contribution of the Greenland Ice Sheet to sea level over the next millennium. *Science Advances*, 5(6), eaav9396. https://doi.org/10.1126/sciadv.aav9396
- Bamber, J. L., Siegert, M. J., Griggs, J., Marshall, S. J., & Spada, G. (2013). Paleofluvial mega-canyon beneath the central Greenland Ice Sheet. Science, 341(6149), 997–999. https://doi.org/10.1126/science.1239794
- Bierman, P. R., Corbett, L. B., Graly, J. A., Neumann, T. A., Lini, A., Crosby, B. T., & Rood, D. H. (2014). Preservation of a preglacial landscape under the center of the Greenland Ice Sheet. Science, 344(6182), 402–405. https://doi.org/10.1126/science.1249047
- Bierman, P. R., Shakun, J. D., Corbett, L. B., Zimmerman, S. R., & Rood, D. H. (2016). A persistent and dynamic East Greenland Ice Sheet over the past 7.5 million years. *Nature*, 540(7632), 256–260. https://doi.org/10.1038/nature20147
- Blard, P.-H., Protin, M., Tison, J.-L., Fripiat, F., Dahl-Jensen, D., Steffensen, J. P., et al. (2023). Basal debris of the NEEM ice core, Greenland: A window into sub-ice-sheet geology, basal ice processes and ice-sheet oscillations. *Journal of Glaciology*, 69(276), 1011–1029. https://doi.org/ 10.1017/jog.2022.122
- Braun, A., Kim, H. R., Csatho, B., & von Frese, R. R. B. (2007). Gravity-inferred crustal thickness of Greenland. Earth and Planetary Science Letters, 262(1–2), 138–158. https://doi.org/10.1016/j.epsl.2007.07.050
- Briner, J. P., Walcott, C. K., Schaefer, J. M., Young, N. E., MacGregor, J. A., Poinar, K., et al. (2022). Drill-site selection for cosmogenic-nuclide exposure dating of the bed of the Greenland Ice Sheet. *The Cryosphere*, 16(10), 3933–3948. https://doi.org/10.5194/tc-16-3933-2022
- Brozena, J. M. (1995). Kinematic GPS and aerogeophysical measurement: Gravity, topography and magnetics. Ph.D. Dissertation. University of Cambridge.
- Campforts, B., Vanacker, V., Herman, F., Vanmaercke, M., Schwanghart, W., Tenorio, G. E., et al. (2020). Parameterization of river incision models requires accounting for environmental heterogeneity: Insights from the tropical Andes. *Earth Surface Dynamics*, 8(2), 447–470. https:// doi.org/10.5194/esurf-8-447-2020
- Christ, A. J., Rittenour, T. M., Bierman, P. R., Keisling, B. A., Knutz, P. C., Thomsen, T. B., et al. (2023). Deglaciation of northwestern Greenland during marine isotope stage 11. Science, 381(6655), 330–335. https://doi.org/10.1126/science.ade4248
- Cochran, J. R., & Bell, R. E. (2016). IceBridge sander AIRGrav L1B geolocated free air gravity anomalies, version 1. NASA National Snow and Ice Data Center Distributed Active Archive Center. https://doi.org/10.5067/R1RQ6NRIJV89
- Cochran, J. R., Burton, B., Frearson, N., & Tinto, K. (2014). IceBridge scintrex CS-3 cesium magnetometer L1B geolocated magnetic anomalies, version 2. NASA National Snow and Ice Data Center Distributed Active Archive Center. https://doi.org/10.5067/OY7C2Y61YSYW
- Cohen, K. M., Finney, S. C., Gibbard, P. L., & Fan, J.-X. (2013). The ICS international chronostratigraphic chart. *Episodes*, 36(3), 199–204. https://doi.org/10.18814/epiiugs/2013/v36i3/002
- Colgan, W., Wansing, A., Mankoff, K., Lösing, M., Hopper, J., Louden, K., et al. (2022). Greenland geothermal heat flow database and map (version 1). Earth System Science Data, 14(5), 2209–2238. https://doi.org/10.5194/essd-14-2209-2022
- Colville, E. J., Carlson, A. E., Beard, B. L., Hatfield, R. G., Stoner, J. S., Reyes, A. V., & Ullman, D. J. (2011). Sr-Nd-Pb isotope evidence for icesheet presence on Southern Greenland during the last interglacial. *Science*, 333(6042), 620–623. https://doi.org/10.1126/science.1204673
- Cooper, M. A., Jordan, T. M., Siegert, M. J., & Bamber, J. L. (2019). Surface expression of basal and englacial features, properties, and processes of the Greenland Ice Sheet. *Geophysical Research Letters*, 46(2), 783–793. https://doi.org/10.1029/2018GL080620
- CReSIS. (2024). Multi-channel radar depth sounder data. Retrieved from https://data.cresis.ku.edu/
- Darbyshire, F. A., Dahl-Jensen, T., Larsen, T. B., Voss, P. H., & Joyal, G. (2018). Crust and uppermost-mantle structure of Greenland and the Northwest Atlantic from Rayleigh wave group velocity tomography. *Geophysical Journal International*, 212(3), 1546–1569. https://doi.org/10. 1093/gji/ggx479
- Dawes, P. R. (2009). The bedrock geology under the inland ice: The next major challenge for Greenland mapping. Geological Survey of Denmark and Greenland Bulletin, 17, 57–60. https://doi.org/10.34194/geusb.v17.5014
- Dawes, P. R., Thomassen, B., & Andersson, T. I. H. (2000). A new volcanic province: Evidence from glacial erratics in western North Greenland. Geological Survey of Denmark and Greenland Bulletin, 186, 35–41. https://doi.org/10.34194/ggub.v186.5213
- Dühnforth, M., Anderson, R. S., Ward, D., & Stock, G. M. (2010). Bedrock fracture control of glacial erosion processes and rates. *Geology*, 38(5), 423–426. https://doi.org/10.1130/G30576.1
- Ekholm, S., Keller, K., Bamber, J. L., & Gogineni, S. P. (1998). Unusual surface morphology from digital elevation models of the Greenland Ice Sheet. *Geophysical Research Letters*, 25(19), 3623–3626. https://doi.org/10.1029/98GL02589
- Fahnestock, M. A., Abdalati, W., Joughin, I., Brozena, J., & Gogineni, P. (2001). High geothermal heat flow, basal melt, and the origin of rapid ice flow in central Greenland. *Science*, 294(5550), 2338–2342. https://doi.org/10.1126/science.1065370
- Fisher, M., Stafford, T., Moon, T., & Thurber, A. (2023). QGreenland (v3.0.0) [Dataset]. Zenodo. https://doi.org/10.5281/zenodo.8326507
  - Forsberg, R., & Jensen, T. (2015). New geoid of Greenland: A case study of terrain and ice effects, GOCE and use of local sea level data. International Association of Geodesy Symposia, 2014, 144–159. https://doi.org/10.1007/1345\_2015\_50
  - Gowan, E. J., Niu, L., Knorr, G., & Lohmann, G. (2019). Geology datasets in North America, Greenland and surrounding areas for use with ice sheet models. *Earth System Science Data*, 11(1), 375–391. https://doi.org/10.5194/essd-11-375-2019
  - Greene, C. A., Gwyther, D. E., & Blankenship, D. D. (2017). Antarctic mapping tools for MATLAB. Computers & Geosciences, 104, 151–157. https://doi.org/10.1016/j.cageo.2016.08.003
  - Gudmundsson, G. H. (2003). Transmission of basal variability to a glacier surface. Journal of Geophysical Research, 108(B5). https://doi.org/10. 1029/2002JB002107
  - Haran, T., Bohlander, J., Scambos, T., Painter, T., & Fahnestock, M. (2013). MEaSUREs MODIS Mosaic of Greenland 2005 (MOG2005) image map, version 1. NASA National Snow and Ice Data Center Distributed Active Archive Center. https://doi.org/10.5067/IAGYM8Q26QRE
  - Harrison, J. C., St-Onge, M. R., Petrov, O. V., Strelnikov, S. I., Lopatin, B. G., Wilson, F. H., et. al. (2011). Geological map of the Arctic. Geological Survey of Canada, "A" Series Map, 2159A. https://doi.org/10.4095/287868
  - Hasterok, D., Halpin, J. A., Collins, A. S., Hand, M., Kreemer, C., Gard, M. G., & Glorie, S. (2022). New maps of global geologic provinces and tectonic plates. *Earth-Science Reviews*, 231, 104069. https://doi.org/10.1016/j.earscirev.2022.104069

- Howat, I., Negrete, A., & Smith, B. (2022). MEaSUREs Greenland Ice Mapping Project (GrIMP) Digital Elevation Model from GeoEye and WorldView imagery, version 2. NASA National Snow and Ice Data Center Distributed Active Archive Center. https://doi.org/10.5067/ BHS4S5GAMFVY
- Jess, S., Peace, A. L., & Schiffer, C. (2020). Sediment supply on the West Greenland passive margin: Redirection of a large pre-glacial drainage system. Journal of the Geological Society, 177(6), 1149–1160. https://doi.org/10.1144/jgs2020-028
- Jones, G. A., Ferreira, A. M. G., Kulessa, B., Schimmel, M., Berbellini, A., & Morello, A. (2021). Uppermost crustal structure regulates the flow of the Greenland Ice Sheet. *Nature Communications*, 12(1), 7307. https://doi.org/10.1038/s41467-021-27537-5
- Jordan, T. A., Thompson, S., Kulessa, B., & Ferraccioli, F. (2023). Geological sketch map and implications for ice flow of Thwaites Glacier, West Antarctica, from integrated aerogeophysical observations. *Science Advances*, 9(22). https://doi.org/10.1126/sciadv.adf2639
- Joughin, I., Smith, B., & Howat, I. (2017). A complete map of Greenland ice velocity derived from satellite data collected over 20 years. *Journal of Glaciology*, *56*(243), 1–11. https://doi.org/10.1017/jog.2017.73
- Joughin, I. R., Smith, B. E., Howat, I. M., & Scambos, T. A. (2016). MEaSUREs multi-year Greenland Ice Sheet velocity mosaic, version 1. NASA National Snow and Ice Data Center Distributed Active Archive Center. https://doi.org/10.5067/QUA5Q9SVMSJG
- Keisling, B. A., Nielsen, L. T., Hvidberg, C. S., Nuterman, R., & DeConto, R. M. (2020). Pliocene–Pleistocene megafloods as a mechanism for Greenlandic megacanyon formation. *Geology*, 48(7), 737–741. https://doi.org/10.1130/g47253.1
- Kirkham, J. D., Hogan, K. A., Larter, R. D., Arnold, N. S., Ely, J. C., Clark, C. D., et al. (2022). Tunnel valley formation beneath deglaciating midlatitude ice sheets: Observations and modelling. *Quaternary Science Reviews*, 323, 107680. https://doi.org/10.1016/j.quascirev.2022.107680
- Lazarus, E. D., & Constantine, J. A. (2013). Generic theory for channel sinuosity. Proceedings of the National Academy of Sciences of the United States of America, 110(21), 8447–8452. https://doi.org/10.1073/pnas.1214074110
- Lea, E. J., Jamieson, S. S. R., & Bentley, M. J. (2023). Alpine topography of the Gamburtsev Subglacial Mountains, Antarctica, mapped from ice sheet surface morphology. *The Cryosphere Discussions*. https://doi.org/10.5194/tc-2023-94
- Li, L., Aitken, A. R. A., Lindsay, M. D., & Kulessa, B. (2022). Sedimentary basins reduce stability of Antarctic ice streams through groundwater feedbacks. *Nature Geoscience*, 15(8), 645–650. https://doi.org/10.1038/s41561-022-00992-5
- Livingstone, S. J., Chu, W., Ely, J. C., & Kingslake, J. (2017). Paleofluvial and subglacial channel networks beneath Humboldt Glacier, Greenland. *Geology*, 45(6), 551–554. https://doi.org/10.1130/g38860.1
- MacGregor, J., Colgan, W. T., Paxman, G. J. G., Tinto, K. J., Csathó, B., Darbyshire, F. A., et al. (2024). Geologic provinces beneath the Greenland Ice Sheet constrained by geophysical data synthesis (v1) [Dataset]. GEUS Dataverse. https://doi.org/10.22008/FK2/BUQQ9C
- MacGregor, J. A., Boisvert, L. N., Medley, B., Petty, A. A., Harbeck, J. P., Bell, R. E., et al. (2021). The scientific legacy of NASA's Operation IceBridge. *Reviews of Geophysics*, 59(2), e2020RG000712. https://doi.org/10.1029/2020RG000712
- MacGregor, J. A., Bottke, W. F., Jr., Fahnestock, M. A., Harbeck, J. P., Kjær, K. H., Paden, J. D., et al. (2019). A possible second large subglacial impact crater in northwest Greenland. *Geophysical Research Letters*, 46(3), 1496–1504. https://doi.org/10.1029/2018GL078126
- MacGregor, J. A., Chu, W., Colgan, W. T., Fahnestock, M. A., Felikson, D., Karlsson, N. B., et al. (2022). GBaTSv2: A revised synthesis of the likely basal thermal state of the Greenland Ice Sheet. *The Cryosphere*, 16, 3033–3049. https://doi.org/10.5194/tc-2022-40
- Maier, N., Gimbert, F., & Gillet-Chaulet, F. (2022). Threshold response to melt drives large-scale bed weakening in Greenland. Nature, 607(7920), 714–720. https://doi.org/10.1038/s41586-022-04927-3
- Martos, Y. M., Jordan, T. A., Catalán, M., Jordan, T. M., Bamber, J. L., & Vaughan, D. G. (2018). Geothermal heat flux reveals the Iceland hotspot track underneath Greenland. *Geophysical Research Letters*, 45(16), 8214–8222. https://doi.org/10.1029/2018GL078289
- Meyer, B., Chulliat, A., & Saltus, R. (2017). Derivation and error analysis of the Earth magnetic anomaly grid at 2 arc min resolution version 3 (EMAG2v3). *Geochemistry, Geophysics, Geosystems, 18*(12), 4522–4547. https://doi.org/10.1002/2017GC007280
- Mordret, A. (2018). Uncovering the Iceland hotspot track beneath Greenland. Journal of Geophysical Research: Solid Earth, 123(6), 4922–4941. https://doi.org/10.1029/2017JB015104
- Morlighem, M., Williams, C. N., Rignot, E., An, L., Arndt, J. E., Bamber, J. L., et al. (2017). BedMachine v3: Complete bed topography and ocean bathymetry mapping of Greenland from multibeam echo sounding combined with mass conservation. *Geophysical Research Letters*, 44(11), 051–11061. https://doi.org/10.1002/2017GL074954
- Morlighem, M., Williams, C. N., Rignot, E., An, L., Arndt, J. E., Bamber, J. L., et al. (2022). IceBridge BedMachine Greenland, version 5. NASA National Snow and Ice Data Center Distributed Active Archive Center. https://doi.org/10.5067/GMEVBWFLWA7X
- Ng, F. S. L., Ignéczi, Á., Sole, A. J., & Livingstone, S. J. (2018). Response of surface topography to basal variability along glacial flowlines. J. Geophys. Res. Earth Surface, 123(10), 2319–2340. https://doi.org/10.1029/2017JF004555
- Nielsen, T. K., Larsen, H. C., & Hopper, J. R. (2002). Contrasting rifted margin styles south of Greenland: Implications for mantle plume dynamics. *Earth and Planetary Science Letters*, 200(3–4), 271–286. https://doi.org/10.1016/s0012-821x(02)00616-7
- Nutman, A. P., Bennett, V. C., Hidaka, H., Henriksen, N., & Ali, S. (2019). The Archean Victoria Fjord terrane of northernmost Greenland and geodynamic interpretation of Precambrian crust in and surrounding the Arctic Ocean. *Journal of Geodynamics*, 129, 3–23. https://doi.org/10. 1016/j.jog.2019.03.006
- Ockenden, H., Bingham, R. G., Curtis, A., & Goldberg, D. (2022). Inverting ice surface elevation and velocity for bed topography and slipperiness beneath Thwaites Glacier. *The Cryosphere*, 16, 3867–3887. https://doi.org/10.5194/tc-2021-287
- Ockenden, H., Bingham, R. G., Curtis, A., & Goldberg, D. (2023). Ice-flow perturbation analysis: A method to estimate ice-sheet bed topography and conditions from surface datasets. *Journal of Glaciology*, 1–10. https://doi.org/10.1017/jog.2023.50
- Paxman, G. J. G. (2023). Patterns of valley incision beneath the Greenland Ice Sheet revealed using automated mapping and classification. *Geomorphology*, 436, 108778. https://doi.org/10.1016/j.geomorph.2023.108778
- Paxman, G. J. G., Austermann, J., & Hollyday, A. (2022a). Total isostatic response to the complete unloading of the Greenland and Antarctic Ice Sheets. Scientific Reports, 12(1), 11399. https://doi.org/10.1038/s41598-022-15440-y
- Paxman, G. J. G., Austermann, J., & Hollyday, A. (2022b). Grid files of the total isostatic response to the complete unloading of the Greenland and Antarctic Ice Sheets (version 2) [Dataset]. Arctic Data Center. https://doi.org/10.18739/A2WS8HN3C
- Paxman, G. J. G., Austermann, J., & Tinto, K. J. (2021a). A fault-bounded palaeo-lake basin preserved beneath the Greenland Ice Sheet. Earth and Planetary Science Letters, 553, 116647. https://doi.org/10.1016/j.epsl.2020.116647
- Paxman, G. J. G., Jamieson, S. S. R., Dolan, A. M., & Bentley, M. J. (2024). Subglacial valleys preserved in the highlands of south and east Greenland record restricted ice extent during past warmer climates. *The Cryosphere*, 18(3), 1467–1493. https://doi.org/10.5194/tc-18-1467-2024
- Paxman, G. J. G., Jamieson, S. S. R., Ferraccioli, F., Jordan, T. A., Bentley, M. J., Ross, N., et al. (2019). Subglacial geology and geomorphology of the Pensacola-pole basin, East Antarctica. *Geochemistry, Geophysics, Geosystems*, 20(6), 2786–2807. https://doi.org/10.1029/ 2018GC008126

- Paxman, G. J. G., Tinto, K. J., & Austermann, J. (2021). Neogene–Quaternary uplift and landscape evolution in northern Greenland recorded by subglacial valley morphology. *Journal of Geophysical Research: Earth Surface*, 126(12), e2021JF006395. https://doi.org/10.1029/ 2021JF006395
- Pedersen, M., Weng, W. L., Keulen, N., & Kokfelt, T. K. (2013). A new seamless digital 1:500,000 scale geological map of Greenland. Geological Survey of Denmark and Greenland Bulletin, 28, 65–68. https://doi.org/10.34194/geusb.v28.4727
- Pourpoint, M., Anandakrishnan, S., Ammon, C. J., & Alley, R. B. (2018). Lithospheric structure of Greenland from ambient noise and earthquake surface wave tomography. *Journal of Geophysical Research: Solid Earth*, 123(9), 1516–1539. https://doi.org/10.1029/2018JB015490
- Rezvanbehbahani, S., Stearns, L. A., Kadivar, A., Walker, J. D., & van der Veen, C. J. (2017). Predicting the geothermal heat flux in Greenland: A machine learning approach. *Geophysical Research Letters*, 44(24), 12271–12279. https://doi.org/10.1002/2017GL075661
- Rogozhina, I., Petrunin, A. G., Vaughan, A. P. M., Steinberger, B., Johnson, J. V., Kaban, M. K., et al. (2016). Melting at the base of the Greenland ice sheet explained by Iceland hotspot history. *Nature Geoscience*, 9(5), 366–369. https://doi.org/10.1038/ngeo2689
- Rose, K. C., Ferraccioli, F., Jamieson, S. S. R., Bell, R. E., Corr, H., Creyts, T. T., et al. (2013). Early East Antarctic ice sheet growth recorded in the landscape of the Gamburtsev subglacial mountains. *Earth and Planetary Science Letters*, 375, 1–12. https://doi.org/10.1016/j.epsl.2013. 03.053
- Rose, K. C., Ross, N., Bingham, R. G., Corr, H. F. J., Ferraccioli, F., Jordan, T. A., et al. (2014). A temperate former West Antarctic ice sheet suggested by an extensive zone of subglacial meltwater channels. *Geology*, 42(11), 971–974. https://doi.org/10.1130/G35980.1
- Ross, N., Jordan, T. A., Bingham, R. G., Corr, H. F. J., Ferraccioli, F., Le Brocq, A., et al. (2014). The Ellsworth subglacial highlands: Inception and retreat of the West Antarctic ice sheet. *Geological Society of America Bulletin*, 126(1–2), 3–15. https://doi.org/10.1130/B30794.1
- Scambos, T. A., & Haran, T. (2002). An image-enhanced DEM of the Greenland ice sheet. Annals of Glaciology, 34, 291–298. https://doi.org/10. 3189/172756402781817969
- Sergienko, O. V. (2012). The effects of transverse bed topography variations in ice-flow models. *Journal of Geophysical Research*, 117(F3), F03011. https://doi.org/10.1029/2011jf002203
- Steinberger, B., Bredow, E., Lebedev, S., Schaeffer, A., & Torsvik, T. H. (2019). Widespread volcanism in the Greenland–North Atlantic region explained by the Iceland plume. *Nature Geoscience*, 12(1), 61–68. https://doi.org/10.1038/s41561-018-0251-0
- Thrane, K. (2002). Relationships between Archaean and Palaeoproterozoic crystalline basement complexes in the southern part of the East Greenland Caledonides: An ion microprobe study. Precambrian Research, 113(1-2), 19-42. https://doi.org/10.1016/S0301-9268(01)00198-X
- Tinto, K. (2020). IceBridge LDEO gravimeter suite L1B geolocated free air gravity anomalies, version 1. NASA National Snow and Ice Data Center Distributed Active Archive Center. https://doi.org/10.5067/SU3MNDAOQ7LZ
- Tinto, K. J., Bell, R. E., Cochran, J. R., & Münchow, A. (2015). Bathymetry in Petermann fjord from operation IceBridge aerogravity. *Earth and Planetary Science Letters*, 422, 58–66. https://doi.org/10.1016/j.epsl.2015.04.009
- Tinto, K. J., Padman, L., Siddoway, C. S., Springer, S. R., Fricker, H. A., Das, I., et al. (2019). Ross Ice Shelf response to climate driven by the tectonic imprint on seafloor bathymetry. *Nature Geoscience*, 12(6), 441–449. https://doi.org/10.1038/s41561-019-0370-2
- Watt, W. S. (2019). Stratigraphic lexicon for Greenland. Geological Survey of Denmark and Greenland, 237. https://doi.org/10.22008/geusbook/ strat-lex-greenland
- Weis, D., Demaiffe, D., Souchez, R., Gow, A. J., & Meese, D. A. (1997). Ice sheet development in Central Greenland: Implications from the Nd, Sr and Pb isotopic compositions of basal material. *Earth and Planetary Science Letters*, 150(1–2), 161–169. https://doi.org/10.1016/S0012-821X(97)00073-3
- White, L. F., Bailey, I., Foster, G. L., Allen, G., Kelley, S. P., Andrews, J. T., et al. (2016). Tracking the provenance of Greenland-sourced, Holocene aged, individual sand-sized ice-rafted debris using the Pb-isotope compositions of feldspars and <sup>40</sup>Ar/<sup>39</sup>Ar ages of hornblendes. *Earth and Planetary Science Letters*, 433, 192–203. https://doi.org/10.1016/j.epsl.2015.10.054
- Winsborrow, M. C. M., Clark, C. D., & Stokes, C. R. (2010). What controls the location of ice streams? Earth-Science Reviews, 103, 45–59. https://doi.org/10.1016/j.earscirev.2010.07.003